Author's response to reviews

Title: The Radiation Issue in Cardiology: the time for action is now

Authors:

Eugenio Picano MD (picano@ifc.cnr.it)
Eliseo Vano PhD (eliseov@med.ucm.es)

Version: 2 Date: 21 October 2011

Author's response to reviews:

Pisa, 21 October 2011

Dear Editor,

below please find our responses to the comments of Referees 1 and 2.
We have fully responded to each point and we hope we have made the appropriate revisions in red in our manuscript. We believe the manuscript is now improved.

Answer to Referee 1
Ref. 57, 66, 78, 79 are incomplete.
We completed the references.

In Tab. 1 the Ref. 24 for endovascular thoraco-abdominal aneurysm repair procedures doesn't match to
Ref. 25 in the text (see page 5, line 18).
We corrected the right references in the text and in the table 1.

Answer to Referee 2
AHA, ACC, ICRP, PCI, MPI, MDCT, IAEA. UNSCEAR, LNT.
We included a list of abbreviations and spelled the abbreviations.

Sincerely

Eugenio Picano, MD, PhD,